Author's response to reviews

Title: Genomic alterations identified by array comparative genomic hybridization as prognostic markers in tamoxifen-treated estrogen receptor-positive breast cancer

Authors:

Wonshik Han (hanw@snu.ac.kr)
Mi-Ryung Han (genetic0309@hanmail.net)
Jason J Kang (jkang@macrogen.com)
Ji-Yeon Bae (jita94@hanmail.net)
Ji Hyun Lee (jhlee@macrogen.com)
Young Ju Bae (joostat@macrogen.com)
Jeong Eon Lee (paojlus@hanmail.net)
Hyunk-Jae Shin (drgss@chollian.net)
Ki-Tae Hwang (kiterius@medimail.co.kr)
Sung-Eun Hwang (hserin@empal.com)
Sung-Won Kim (brca@snubh.org)
Dong-Young Noh (dynoh@plaza.snu.ac.kr)

Version: 3 Date: 2 February 2006

Author's response to reviews: see over
Dear Editor,

Thank you for the decision to accept our manuscript for publication. Please find enclosed the revised version entitled, “Genomic alterations identified by array comparative genomic hybridization as prognostic markers in tamoxifen-treated estrogen receptor-positive breast cancer”.

The reviewer’s and editor’s comments have been addressed as follows:

**Reviewer 1 (A. Dellas):**

Comment 1: We changed the term “gene copy number” to “DNA copy number” in page 4 and 8.

Comment 2: The term “genes” was replaced by “chromosomal location” in page 9.

Comment 3: The sentence in page 17 was corrected.

**Editor’s comments:**

Comment 1: All underlines were removed for the manuscript.

Comment 2: The name of IRB was documented in page 4, line 24.

Comment 3: The reference list was corrected showing all authors’ names.
Sincerely yours,

Dong-Young Noh, M.D., PhD
Department of Surgery
Seoul National University College of Medicine
28 Yongon-dong, Chongno-gu
Seoul, 110-744 KOREA
Phone: +82-2-760-2318; Fax +82-2-766-3975
Email: dynoh@plaza.snu.ac.kr